Author's response to reviews

Title: Are osteoporotic fractures being adequately investigated?: A questionnaire of GP & orthopaedic surgeons

Authors:

George MR Chami (georgechami@doctors.org.uk)
Lee MR Jeys (lee.jeys@btclick.com)
Mathew MR Freudmann (m@freudmann.co.uk)
Louise DR Connor (louise_connor@hotmail.com)
Mashood DR Siddiqi (masiddiqi@doctors.org.uk)

Version: 4 Date: 20 December 2005

Author's response to reviews:

Dear Mr Austin,

I resubmitted the revised manuscript as the table has been split between two pages. This formatting error has been corrected in this submission. No other changes have been done.

Kind Regards

George Chami